Author's response to reviews

Title: Elevated preoperative peripheral blood monocyte count predicts poor prognosis for hepatocellular carcinoma after hepatic resection

Authors:

Shun-Li Shen (shunlishen@163.com)
Shun-Jun Fu (fsj103@163.com)
Xiong-Qing Huang (hxiongqing@21cn.com)
Bin Chen (chenbin241@sohu.com)
Ming Kuang (kuangm@mail.sysu.edu.cn)
Shao-Qiang Li (lisq@medmail.com.cn)
Yun-Peng Hua (hyp0427@163.com)
Li-Jian Liang (lianglj@medmail.com.cn)
Baogang Peng (pengbaogang@medmail.com.cn)

Version: 3 Date: 31 July 2014

Author's response to reviews: see over
Dear Annie Lyn Bravo

I'm very glad to receive your letter.

We have used editing service from Edanz (C1407-35817-ShenShunli) for the revision of our manuscript as recommended. In addition, we have removed all word contractions ('haven't' and 'didn't') from the entire text of the manuscript.

Thank you very much for your help.

Sincerely,

Bao-Gang Peng, MD, PhD

Department of Hepatobiliary Surgery

The First Affiliated Hospital of Sun Yat-Sen University

58, Zhongshan Road 2, Guangzhou 510080, China.

Email: pengbaogang@medmail.com.cn

Telephone: +86-20-87755766-8214

Fax: +86-20-87755766-8663